Author's response to reviews

Title: Scanning for satisfaction or digging for dismay? Comparing findings from a postal survey with those from a focus group-study

Authors:

Benedicte Carlsen (benedicte.carlsen@uni.no)
Claire Glenton (claire.glenton@kunnskapssenteret.no)

Version: 3 Date: 28 August 2012

Author's response to reviews:

Dear Editors,

We were pleased to receive your acceptance for publication of our manuscript. As requested we have now reformatted the manuscript as according to the "debate" template. This did not entail substantial changes as "debatee" was the format we had originally intended the manuscript to suit. We have thus altered some of the headings in the abstract and the main text, and we have added a short summary section at the end of the main text.

Hopefully, the text is now ready for publication.

Bw,

Benedicte Carlsen and Claire Glenton